

Using Budget Standards to Assess the Well-Being of Families

Author/Contributor: Saunders, Peter

Publication details:

Working Paper No. 93 SPRC Discussion Paper 0733416853 (ISBN) 1447-8978 (ISSN)

Publication Date: 1998

DOI: https://doi.org/10.26190/unsworks/225

License:

https://creativecommons.org/licenses/by-nc-nd/3.0/au/ Link to license to see what you are allowed to do with this resource.

Downloaded from http://hdl.handle.net/1959.4/34047 in https:// unsworks.unsw.edu.au on 2023-03-31



USING BUDGET STANDARDS TO ASSESS THE WELL-BEING OF FAMILIES

by Peter Saunders

SPRC Discussion Paper No. 93 December 1998

ISSN 1037 2741 ISBN 7334 1685 3

This paper was presented at the 6th Australian Institute of Family Studies Conference, *Changing Families, Challenging Futures*, Melbourne, 25-27 November 1998. The author acknowledges the contributions of his colleagues Michael Bittman, Bruce Bradbury, Jenny Chalmers and Marilyn McHugh to the budget standards research on which this paper is based. Assistance with the statistical analysis was provided by George Matheson. The usual caveats apply. The Social Policy Research Centre (formerly the Social Welfare Research Centre) was established in January 1980 under an Agreement between the University of New South Wales and the Commonwealth Government. In accordance with the Agreement the Centre is operated by the University as an independent unit within the University. The Director of the Centre is responsible to the Vice-Chancellor and receives advice in formulating the Centre's research agenda from a Management Board.

SOCIAL POLICY RESEARCH CENTRE DISCUSSION PAPERS are intended as a forum for the publication of selected research papers on research within the Centre, or commissioned by the Centre, for discussion and comment in the research community and/or welfare sector prior to more formal publication. Limited copies of each DISCUSSION PAPER will be available on a first-come, first-served basis from the Publications Officer, Social Policy Research Centre, University of New South Wales, Sydney NSW 2052 [tel: (02) 9385 7800]. A full list of DISCUSSION PAPERS can be found at the back of this DISCUSSION PAPER.

The series is indebted to Diana Encel for her continuing editorial contribution.

As with all of the Centre's publications, the views expressed in this DISCUSSION PAPER do not reflect any official position on the part of the Centre.

Tony Eardley Editor

Abstract

This paper describes the methods used by the Budget Standards Unit at the Social Policy Research Centre to develop a set of indicative budget standards for a range of Australian households. Some of the results from the project are then compared with estimates of actual household expenditures derived from the Household Expenditure Survey conducted by the Australian Bureau of Statistics. The sensitivity of budget standards to some of the key assumptions and judgements made in developing them is illustrated in two examples: housing costs and spatial variations in prices. The paper concludes that one important contribution that budget standards research can make to discussion of the adequacy of household incomes is in providing a transparent framework for selecting items needed to maintain a particular standard of living and translating them through prices into the budgets required to purchase them.

1 Introduction

The release earlier this year of the results from the Social Policy Research Centre (SPRC) budget standards study represents the latest chapter in a long line of Australian research on budget standards. Over ninety years ago, Justice Higgins relied on a rudimentary budget standard to assist in the determination of the basic wage in his 'Harvester Judgement' - a decision which was to influence the course of working family incomes until the basic wage itself was abandoned in 1967. Thereafter, despite further development of the topic in the 1920s by A.B.Piddington in his work as the Chairman of the Royal Commission on the Basic Wage and subsequently, as well as by Professor Wilfred Prest at the University of Melbourne in the war years, budget standards have fallen out of favour for much of the postwar period.

In the early 1980s, for example, the Social Welfare Policy Secretariat argued in its *Report on Poverty Measurement* (1981: 39) that there was a need for 'debate about the nature and minimum amounts of goods and services which may comprise an austere but tolerable standard of living', but saw severe practical and conceptual difficulties with the budget standards approach based in part on evidence that those working in welfare agencies seemed unable to agree on what constitutes a minimum standard of living in modern Australian society. In developing its budget standards, the Budget Standards Unit (BSU) research team became all too aware of these practical and conceptual difficulties, but the Budget Standards Report (Saunders et al., 1998) represents an attempt to overcome them in ways which produce new insights into the lifestyles and living circumstances of Australian families at the end of the 1990s.

How well this task has been achieved will be for others to judge. If people find the BSU budget standards useful, then the judgements and assumptions that have been made in developing them will either receive endorsement or be modified. Either way, the standards themselves will gain legitimacy. Some will argue that it is unreasonable to expect those who wish to use a budget standard to be expected to go to the trouble of understanding all of the judgements, limitations and qualifications that apply to them. Indeed, many critics of the budget standards method see the tendency to treat a budget standard as an objective statistic, rather than a normative indicator, as one of its most serious weaknesses - and its greatest danger.

The BSU response to this kind of criticism has been to argue that our budget standards are indicative and intended to generate debate about our methods, assumptions and judgements. The fact that these judgements and assumptions are made explicit in our Report provides a basis for others to debate their relevance. My view is that for a budget standard to be *useful*, it must be *used* and this will only occur when potential users have a degree of *understanding* of how the standard was derived and a degree of agreement with what it represents. The best way to understand how a budget standard was developed and what it means is to use it.

This paper contributes to that process by describing how the BSU budget standards were developed, by comparing the estimates themselves with alternative indicators of well-being and by exploring some aspects of the sensitivity of the standards to price variations. The paper is organised as follows: Section 2 introduces the budget standards concept and explains in general terms how the BSU standards were developed. Section 3 presents some of the results from the budget standards project and compares the estimates with actual household expenditures derived from survey data collected by the Australian Bureau of Statistics (ABS) in its *Household Expenditure Survey* (HES). Section 4 discusses the sensitivity of the estimated budget standards in two illustrative dimensions: housing costs and spatial variations in prices. Section 5 summarises the main conclusions.

2 Explaining the BSU Approach

General Considerations

A budget standard estimates what is needed, in terms of material goods and services, by a particular type of family in order to achieve a particular standard of living in a particular place at a particular time. The definition is important in three regards. First, it emphasises that attention focuses on the *material dimensions of well-being* rather than its psychological or subjective determinants. Secondly, the definition draws attention to the *specificity* of

any measure of the standard of living, in the sense that it depends not only on how family circumstances influence family needs and what this implies for what items to include - the 'what' issue, but also on the context within which these needs are met - the 'where' and 'when' aspects. Perhaps most significant of all, the definition highlights the fact that, in principle at least, it is possible to develop a budget standard that corresponds to *any standard of living*.

The budget standards method begins by specifying the needs that have to be met in order to maintain a given standard of living, then identifies the goods and services required to meet these needs, and finally costs them and adds them up to produce a total budget. If provided with this amount of resources (either as cash or perhaps as access to the identified services free of charge) a family of a given type will have the *opportunity*, through consumption of goods and services, to achieve the standard of living to which the budget itself corresponds. (This statement ignores the impact of income tax which, if it would reduce the income corresponding to a derived budget standard, would need to be added to the budget to arrive at the gross income required to purchase the budget out of after-tax income). Whether the family actually chooses the same or a similar or an entirely different pattern of consumption will depend upon the tastes and preferences of family members. However, if a family is denied the resources that correspond to a particular budget standard, than it will not be able to meet all of the needs on which that standard is based.

This last observation explains why budget standards have had a long association with research on poverty, where poverty is defined as a level of resources adequate to sustain a minimum standard of living in which only basic needs - physical *and* social - can be met. If this poverty standard is socially endorsed as an adequacy benchmark, then the circumstances of families with insufficient resources to allow them to maintain the standard is a cause for social concern and provides a rationale for social policy intervention. The nature of such intervention will depend upon the specification of the poverty standard and there may be disagreement over this, but such debate is independent of how the budget standards themselves

are developed. To estimate a budget standard is one thing; to use it as a benchmark to measure poverty or income adequacy generally is another.

Development of a measure of the standard of living which allows the position of different families to be compared is one of the most challenging tasks confronting social policy research. Although the notion of standard of living features prominently in discussions of social trends and social policies, there are many difficult issues associated with defining and measuring the standard of living. The conceptual problems were highlighted some years ago by this year's Nobel Laureate in Economics - Professor Amartya Sen - in his Tanner Lectures presented in 1985 at the University of Cambridge (Sen, 1988), which he began by observing that:

While people are not prone to ask each other, 'How is your standard of living these days?' (at least, not yet), we do not believe we are indulging in technicalities when we talk about the living standard of the pensioners, or of the nurses, or of the miners, or - for that matter - of the chairman of the Coal Board. The standard of living communicates, and does so with apparent ease. And yet the idea is full of contrasts, conflicts and even contradictions. Within the general notion of the standard of living, divergent and rival views of the goodness of life co-exist in an unsorted bundle. (Sen, 1988: 1)

Later in these Lectures, Sen explains the difference between a person's standard of living and their well-being, the latter (broader) notion making no qualifications as to the nature of the person's life, whereas the former makes 'exactly that qualification' (Sen, 1988: 28). Sen argues that whereas undertaking 'sympathetic' tasks such as helping others may make a person feel and actually be better off, they do not add to their standard of living. In moving from well-being to living standard, it is necessary to adopt a narrower focus which ignores influences on one's well-being that arise from outside of one's own life. It is well-being in this internalised, objective

sense that corresponds to the notion of standard of living that a budget standard attempts to capture.

Some of the practical problems involved in defining and measuring the standard of living in an Australian context were identified by Brownlee (1990) in a literature review conducted in the early stages of the Australian Living Standards Study undertaken by the Australian Institute of Family Studies (AIFS). That study decided to use a multi-dimensional framework covering living conditions in fourteen spheres of life including, in addition to economic resources, health, education, recreation, security, social and political participation, access to information, family relationships and personal well-being. Such an approach provides considerable flexibility, but begs the question of how to combine achievements in each of the different spheres into a single measure. The key issue here, as Travers and Richardson (1993: 21) observe, is what weights to use when combining each item. Although this problem can be avoided by sticking with 14 separate indicators, this does not allow the living standards of different groups to be compared and thus provides only limited information for research and policy purposes.

Unlike the AIFS approach, most studies of living standards have adopted a single measure, generally one that incorporates two dimensions: the amount of *resources* available and the *needs* that have to be met. The measure of economic well-being preferred by most economists uses disposable income to reflect the level of available resources, with adjustment by an equivalence scale used to capture the needs of different families. The resulting measure of equivalent disposable income forms the basis of the Henderson poverty framework and is widely used in empirical studies of poverty and income distribution (Saunders, 1997a). Even within this narrow paradigm, some have argued for the use of resource measures other than income (e.g. expenditure - see Saunders, 1997b), or for income to be adjusted to reflect wealth holdings (as is done, in a simplified manner in the after-housing costs poverty line).

In contrast, the budget standards approach treats resources and needs in a closely related but integrated manner. Thus, rather than adjusting resources

by the use of an equivalence scale that has been derived independently to provide a measure of the standard of living, the budget standards approach identifies and costs needs by developing a basket of goods and services which, when priced, indicates the level of resources required to purchase the items that satisfy those needs. This seems a preferable approach, in part because it takes needs - and the cost of meeting them - as the starting point for measuring the standard of living, but also because it makes it explicit that additional needs will involve additional resources if the standard of living is to remain constant. Furthermore, the budget standards approach makes it possible to assess the resource cost of meeting *specific* needs (e.g. those associated with working or looking for work, or due to the presence of children) in order for an economic unit such as the family to maintain the same standard of living. This appears to be the sense in which Sen regards the notion of living standard as one which communicates with ease.

The BSU project was set the task of developing a budget standard for a broad range of household types at two separate standards: a modest but adequate standard and a low cost standard. (Each household was assumed to contain either single people or nuclear families, and so the term 'family' can be used interchangeably with 'household'). The modest but adequate standard is one which affords full opportunity to participate in contemporary Australian society and the basic options it offers, lying between the standards of survival and decency and those of luxury. It attempts to describe the situation of households whose standard of living falls somewhere around the median standard experienced in the Australian community as a whole. The low cost standard, in contrast, is seen as one which may require frugal and careful management of resources but still allow social and economic participation consistent with community standards and enable the individual to fulfil community expectations in the workplace, at home and in the community. Whilst not seen as a minimum standard, the low cost standard is one below which it would become increasingly difficult to maintain an acceptable standard of living because of the increased risk of deprivation and disadvantage. In round terms, the low cost standard corresponds to what is achievable at about one-half of the median standard in the community.

These two concepts have evolved from the budget standards research now conducted in an increasing number of countries. Most of the industrial countries that have developed a budget standard (of which there is now a long list, including Canada, Denmark, Finland, Germany, Sweden, the UK and the US) have adopted a variant of one or both of these concepts, the original specifications of which were initially developed in research undertaken by the US Department of Labour over 70 years ago and refined in the 1980s by the work of the Expert Committee on Family Budget Revisions chaired by Harold Watts (Watts, 1980).

A budget standard must incorporate both normative and behavioural factors. The former may have an official or quasi-official status if they take the form of official guidelines published by the relevant authorities. Many countries, for example, have nutritional guidelines developed and endorsed by such bodies as the National Health and Medical Research Council (NH&MRC) or its equivalent and these can be used to develop a food budget. In other areas, where there are no established social norms available, budget standards are based on expert recommendations which have no official status. The BSU housing budgets, for example, are based on a specification of housing needs derived using a normative occupancy standard which specifies the number of bedrooms required to house households of differing size and composition.

These normative standards must also to some extent reflect the actual behavioural patterns of the population if their relevance is not to be severely circumscribed. In the area of food, for example, a diet consisting mainly of lentils and brown rice may meet the NH&MRC dietary guidelines, but be of little relevance to the actual eating habits of the vast majority of Australians. It is thus often necessary to modify budgets derived directly from the existing normative standards by using behavioural data that ground them in the reality of everyday experience and custom. The difficulty is how this can be achieved without undermining the ability of a budget standard to reflect normative judgments about *needs*, as opposed to the resource *constraints* that also influence actual patterns of behaviour.

Since one of the main uses to which a budget standard can be put is to provide an *independent* benchmark for assessing the adequacy of incomes and standards of living, it is important that the standards themselves do not mirror the effects of the constraints under which different families operate. A budget standard must thus attempt to identify and cost the *needs* of families in ways that are independent of how effectively these needs are being met within current resource constraints, but also be responsive to patterns of behaviour that reflect both needs *and* constraints.

Although there are dangers in relying too heavily on behavioural data when developing a budget standard, there are many areas where this is the only practical option. In the case of clothing and footwear or household furnishings and services, for example, there are few available social norms to guide development of a budget standard. In these instances, all that can be done is to incorporate those which do exist (e.g. in relation to accepted standards of workmanship and materials), but to base the budgets on informed judgments regarding what kinds of items correspond to each standard, what their quality should be, how long they are likely to last and what their price is.

This inevitably means that the purely normative basis for a budget standard is compromised. But this does not make a budget standard any more arbitrary than any other living standard or poverty indicator. As US poverty expert Patricia Ruggles has argued:

> ... even though there is no one 'right' bundle of consumption needs for the poor that all experts would agree on, we do know enough to eliminate a very large number of clearly wrong answers. In this sense, an expert-determined market basket need not be seen as essentially arbitrary, even conceding that an exact 'scientific' determination of needs is not really possible. (Ruggles, 1990: 49)

In practice, the best that can probably be hoped for from a budget standard is that it incorporates those community norms that are in existence but is also informed by expert judgments and existing patterns of behaviour, to the extent that these are known from available data.

The fact that a budget standard runs into criticism as a measure of what it costs to achieve a particular standard of living reflects the inherent difficulty of that task. Once estimated, a budget standard provides not only a monetary measure of equivalence - how much different households require to meet their needs at the same standard of living - but also allows the actual difference between the living standards of different households to be quantified in statements like: 'Household A's resources, relative to its needs, means that its standard of living is X per cent (or \$Y) above that of Household B, given its resources and needs'. Given the enormous conceptual difficulties surrounding the notion of living standard identified by writers like Sen, it is no surprise that a budget standard *like any alternative measure* is subject to controversy and debate.

The BSU Approach

It is not appropriate to explore here in detail how the BSU budget standards were developed, or to outline the many assumptions, judgments and data on which they are based. These details are spelt out in the 630 pages of the Budget Standards Report, to which interested readers are referred (Saunders et al., 1998). It is, however, worth emphasising that the budget standards estimates were exposed to external review at several stages of their development, including by a Steering Committee comprised of experts in nutrition, health economics, housing, clothing needs, consumer behaviour, social security, the measurement of living standards and family budgeting. The preliminary estimates were also presented to a series of focus groups who provided valuable advice on how the standards related to their own experiences and values and identified areas where revisions were necessary. Finally, the estimates were compared with data on actual expenditure patterns using data from the *1993-94 Household Expenditure Survey* (ABS, 1995a and b), an aspect addressed in more detail later.

In specifying and costing the BSU budgets, and in differentiating between the low cost and modest but adequate standards, the total budget for each household was split into nine main budget areas: housing, energy, food, clothing and footwear, household goods and services, health, transport, leisure and personal care. In areas such as food, clothing and footwear, health and personal care, the budgets were mainly developed separately for each individual and then aggregated to the household level. In contrast, in areas such as housing, energy and transport, the unit of analysis was effectively the household, as it was only possible to specify needs and identify items at this level because most items are consumed jointly by members of the household. In between these two extremes are many items of furniture and consumer durables which meet both individual and household needs.

A budget standard was derived at one or both of the low cost and modest but adequate standards for a range of households that varied according to overall size, the age and gender of individual members, the labour force status of adults and the housing tenure of the household. Not all of these factors can be set independently of each other. Thus, the low cost standards generally assume that adult household members are either unemployed or not in the labour force, whereas the modest but adequate standards assume that at least one (working-age) adult is in full-time employment. In general, the low cost standards apply to households in (private or public) rental accommodation, whereas many of the modest but adequate budgets assumed that the household is a purchaser. (Most of the standards for older households assume that they own their homes outright). In total, 46 separate budgets were developed and costed: 26 at the modest but adequate standard and 20 at the low cost standard.

In determining which items (goods, services and activities) to include in the budgets, an 'ownership rule' was applied, under which only those items owned, services used or activities undertaken by at least 50 per cent of households were included in the modest but adequate budgets. The low cost standards were based on a corresponding 75 per cent rule, this being used in effect to identify which items, services and activities are 'necessities'.

Application of this rule can, in some circumstances lead to an upward bias in the budget standards because of how it is applied in practice. Consider, for example, a situation in which 30 per cent of all families own both a stereo system and a home computer, with the remaining 70 per cent choosing equally between one or the other, but not both. In practice, 65 per cent of the population will own a stereo system (30 per cent plus half of 70 per cent) and 65 per cent will own a home computer, so that application of the budget standard ownership rule would result in *both* items being included (at the modest but adequate standard), since both satisfy the 50 per cent rule. Yet the majority of the population (70 per cent) in fact *choose between* the two items rather than owning both.

This example illustrates the difficulty of dealing with *substitute goods* (which meet similar needs) in developing a budget standard by observing actual ownership patterns. In practice, irrespective of the resources available, all families make choices between goods that satisfy similar needs, yet no one has yet devised a satisfactory method for taking account of this within the budget standards framework.

Another means by which the low cost and modest but adequate standards were differentiated was by using different prices to cost the budgets. In general, the low cost budgets were priced using generic ('No Frills') brands, whereas 'leading brand' items were identified and priced in the modest but adequate budgets. Where there was a range of items that serve the same purpose (e.g. in the case of many larger items of household furniture), a price at the lower end of the range of observed shelf prices in leading retail stores was used when developing the low cost budgets, while something closer to the median price was used in the modest but adequate budgets. Rent levels for renter households at the modest but adequate standard were based on the median rent in the selected location, while the corresponding low cost rents were set at the 25th percentile of the distribution: a procedure similar to that employed in recent US budget standards research (Renwick, 1993; Renwick and Bergmann, 1993).

The budgets apply to households living in Sydney and were costed using February 1997 prices. Wherever possible, items were identified and priced at leading retail outlets (so as to make it easier to apply the budgets to other areas or to re-price them in Sydney at a later date). Families were assumed to be living in the Hurstville Local Government Area (LGA) and house prices and rent levels were those applying in that area. Although the Hurstville LGA is reasonably representative of other LGAs in the Sydney metropolitan region in terms of its demographic and socio-economic profiles, the same cannot be claimed for its representativeness in relation to other parts of the country, and this needs to be kept in mind when assessing the results, particularly when they include housing costs (see below).

The above discussion has attempted to explain briefly how the BSU budget standards were developed and to highlight some of their limitations. In focusing attention on the latter, it is all too easy to convey the impression that the estimates require so many qualifications as to make them virtually useless. In the light of this danger, it is worth concluding this section by reasserting some of the *strengths* of a budget standard. The most important of these is that the method identifies what needs have to be met in order to maintain a given standard of living and then costs them. This is a complex and formidable task, but one that has to be confronted in order to put a material dimension and monetary figure on a particular standard of living. The fact that this requires judgments to be made which many will dispute reflects the inherent difficulties associated with obtaining quantitative measures of the standard of living, rather than any fundamental objection to the notion of a budget standard itself.

As Sen also pointed out in his Tanner Lectures (1988), the concept of a budget standard has been endorsed (even if the term itself was not used) by one of the founding fathers of modern welfare economics, Professor Pigou, who Sen quotes as arguing that:

... a national minimum standard of real income ... must be conceived, not as a subjective minimum of satisfaction, but as an objective minimum of conditions ... [which] includes some defined quantity and quality of house accommodation, of medical care, of education, of food, of leisure, of the apparatus of sanitary convenience and safety where work is carried on, and so on. (Quoted in Sen, 1988: 14) And Sen himself considered that identifying someone with a low standard of living in terms of their (lack of) possession of vital commodities seems 'fair enough', although he went on to argue that the approach needs to be taken further to consider the relationship between *functioning* - the various living conditions that can or cannot be achieved, and *capability* - the ability to achieve them (Sen, 1988: 20-38).

To summarise, budget standard estimates are controversial because, by attempting to put a monetary figure on the standard of living, they confront a series of conceptual (not to mention practical) problems that have occupied some of the finest economists and other social scientists, from Adam Smith to Alfred Marshall, Karl Marx and Pigou to Sen and his contemporaries. It is thus not only appropriate to add a note of caution to the estimates, but also to assess how sensitive they are to changes in the methods used to derive them. This was the strategy adopted by the BSU in its research, and some further illustrations of this approach are presented later.

3 Results

The BSU Budget Standards

Table 1 presents a sample of the budget standards estimates for a range of households, at both the low cost and modest but adequate standards. The standards shown in Table 1 are a sub-set of all those developed and apply to households living in (unfurnished) private rental accommodation (except for the older families, who are assumed to be outright owners). The low cost standards range from \$215 a week for a single woman aged 70 (\$294 a week if she is aged 35), up to \$732 a week for a couple with four children. The corresponding figures at the modest but adequate standard range from \$280 a week for the 70 year-old woman (\$383 a week for the younger woman) to \$1083 a week for the couple with four children. The estimates provide evidence of considerable economies of scale in larger households. Thus, the budget for a couple with four children is less than double that of the couple

Household Type ^(a)	Low Cost Standard (LC) (\$)	Modest but Adequate Standard (MBA) (\$)	Ratio of LC to MBA (%)
Simple formula (F25)	204.0	292.4	767
Single female (F35)	294.0	383.4	/6./
Couple (M40, F35)	381.6	513.8	74.3
Couple (as above) plus one child (G6)	487.2	608.0	80.1
Couple plus two children (G6, B14)	602.1	817.4	73.7
Couple plus three children (G3, G6, B14)	659.3	977.5	67.4
Couple plus four children (G3, G6, B10, B14)	731.8	1082.7	67.6
Sole parent (F35) plus one child (G6)	371.8	519.8	71.5
Sole parent plus two children (G6, B10)	485.7	690.9	70.3
Older female (F70) ^(b)	215.0	280.1	76.8
Older couple (M70, F70) ^(b)	295.6	387.6	76.3

 Table 1: The BSU Budget Standards for Different Private Renter Households:

 February 1997 (Dollars per week)

Notes: a) F35 = female aged 35; M40 = male aged 40, G6 = girl aged 6, B14 = boy aged 14, and so on.

b) Older households are assumed to be outright owners.

Source: Saunders et al., 1998, Tables 12.2, 12.3 and 12.4.

at the low cost standard, and only slightly more than double the corresponding modest but adequate standard. There are also very considerable economies of scale for couples compared with single people.

The final column of Table 1 expresses the low cost standards for each family type as a proportion of the modest but adequate standards for the same family. In the majority of cases, this ratio generally lies between 70 per cent and 75 per cent, yet the definitions of the low cost and modest but adequate standards presented earlier imply that the former should be around 50 per cent of the latter, falling 'about one-half of the median standard in the community', compared with a standard designed to 'fall somewhere around the median'. Table 1 thus suggests that either the low cost standards are somewhat too high, or the modest but adequate standards are somewhat too low (or some combination of the two).

One possible explanation for this which was discussed in the Budget Standards Report (Saunders et al., 1988: 442) is that the notion of the low cost standard is somewhat flawed, in that it may not be possible to specify a budget which *simultaneously* allows 'social and economic participation consistent with community standards' whilst falling 'around one-half of the median standard of living in the community'. Since the budget standards were developed to allow appropriate participation and only compared later with estimates of the median (whose source is explained below), there would seem to be an element of truth in this explanation. This does not mean that the low cost standards are too high. Rather, it suggests that the expectation that one can engage in social participation at a living standard only around half of the median is overly optimistic.

Alternatively, to reduce the low cost budget standards from around 75 per cent to 50 per cent of the modest but adequate standards would require reducing them by about one-third. A cut of this magnitude would have very severe consequences for the standard of living that could be maintained on a low cost budget. It would, for example, drastically reduce the amount of social participation, possibly to the point where the household would be either experiencing social exclusion, or be at great risk of being marginalised.

To take an example from Table 1, to reduce the low cost standard for a couple with two children to one-half of the corresponding modest but adequate standard would involve reducing it from \$602.10 to \$408.70, or by \$193.40. For this family type, a budget of \$408.70 would not allow them to meet their housing, energy, food and clothing needs at the low cost standard - these four areas being costed at over \$416 in the BSU research (Saunders et al., 1998, Table 12.3). There would thus be no money left over for health, transport, leisure and personal care needs - all of which facilitate social participation - nor would there be any money for household goods and services, a category which includes expenditure on many basic household goods as well as education and child care costs. The example illustrates the implications of fixing a low cost budget at around one-half of the modest but adequate budgets shown in Table 1. (In fact, evidence presented later, in Table 3, suggests that the modest but adequate budgets may themselves be

somewhat on the high side, which would make the consequences of setting the low cost standards at half the modest but adequate standard even more serious).

Budget Standards and Living Standards

How do the BSU budget standards compare with the actual standards of living of Australian households? This is an obvious question to pose, and its reply provides further insight into both the level and distribution of actual household material circumstances and the relevance and applicability of the budget standards. However, before such comparisons can be undertaken, it is necessary to ensure consistency between the budget standard estimates and the data with which they are compared. This process raises several important issues that are now briefly discussed.

The data used to compare with the budget standards were taken from the *1993-94 Household Expenditure Survey* (HES) undertaken by the Australian Bureau of Statistics (ABS, 1995a). As noted earlier, the HES data were used as one way of providing a 'reality check' on the preliminary budget standard estimates, although this validation process focused primarily on the *composition* of the standards rather than their overall level and did not thus involve the kind of detailed comparisons that are reported below.

Before the comparisons could be made, it was necessary to make several adjustments to the raw HES data. The first of these involved updating the HES data to February 1997. This was achieved by applying an indexation adjustment equal to the rise in the Consumer Price Index (CPI) between the financial year 1993-94 and the March Quarter of 1997. All negative HES expenditures (which correspond to the revenue earned by selling items) were re-coded to zero so as to conform with the budget standards methodology. The next problem that arose was that the BSU household categories are very precisely defined, particularly in terms of the age, gender and labour force status of individual household members. To restrict the HES data in the same way would result in a sample that is too small to allow meaningful analysis to be undertaken.

It was thus decided to identify a series of HES household categories that, whilst similar to those used in the BSU research, were broader in scope so as to increase the useable sample size. The first step was to restrict the HES sample to only single income unit households, thus corresponding to those specified in the budget standards project. This involved adopting a broader definition of age, so that non-aged adults were defined to include all women under 60 and men under 65, aged adults were all women over 60 and men over 65, and children were defined as being under 15 (with no differentiation by gender). No allowance was made for variations in labour force status, although the BSU categories of household size and structure were replicated. These restrictions mean, for example, that the fourth BSU category shown in Table 1 (the couple consisting of a 40 year-old man, a 35 year-old woman, a 6 year-old girl and a 14 year-old boy) was approximated in the HES data by all two-child couples not living with other people, where the man was under 65, the woman under 60, with two children under 15. Similar approximations were made to all of the other detailed BSU categories shown in Table 1.

The next issue concerns the fact that many items that are recorded in the HES were excluded when the budget standards were being developed. Sometimes, this was for explicitly normative reasons. For example, the budget standards make no allowance for the consumption of tobacco products, for health reasons. Should expenditure on tobacco be excluded from the HES before making the comparisons? Although it can be argued that it should be, so as to achieve as much consistency of coverage and definition as possible, an alternative argument is that if people choose to spend part of their budget on tobacco products, to deduct this expenditure from their resources before comparing the remainder with the budget standards could distort the comparisons. Thus, a household whose total expenditure is above the low cost standard has adequate resources to achieve that standard, irrespective of whether or not it chooses to spend part of those resources on tobacco. A similar line of reasoning can be applied to other items of expenditure such as overseas holidays, that are also excluded from the budget standard calculations. To exclude such items from expenditure would produce an underestimate of the resources available,

potentially leading to an overestimate of the numbers falling below the low cost and/or modest but adequate budget standards. In the light of these considerations, the HES data refer to total household commodity and service expenditure as reported in ABS publications based on the HES (e.g. ABS, 1995a).

One area where problems arise is in the treatment of expenditure on consumer durables. In collecting and reporting the HES data, the ABS adopts an acquisitions approach under which the full cost payable by each household in acquiring a good or service over the survey period, regardless of whether the household actually paid for or consumed the good or service within the period is reported as expenditure on each item (see ABS, 1995a: 3-4 for an explanation of the acquisitions approach). In contrast, as noted earlier, the budget standards approach to the treatment of the cost of consumer durables is to spread the purchase price over the assumed lifetime of each item. This difference will obviously affect comparisons between the two measures.

Consider, for example, a television set costing \$650 that is assumed to last for 10 years. The BSU budgets will include an item corresponding to the cost of a television set equal to \$65 a year or \$1.25 a week. If, Australian households do in fact keep their television sets for 10 years, and if every household owns one, then everyone meets the 'television standard' implicit in the budget standard. However, the HES will only record the expenditures of those who acquired a new television set during the survey period. If this period is one year, then only 10 per cent of households will, on average, record a positive *expenditure* on television sets, even though television *ownership* will, by definition, be equal to 100 per cent. The remaining 90 per cent of the population will have zero recorded expenditure on television sets over the past year, not because they do not own one, but because the set that they do own does not yet need to be replaced. Yet if expenditure on televisions as recorded in the HES is compared with a budget standard, 90 per cent will apparently fall below this implied television standard.

Such problems will multiply in practice as each consumer durable item is considered, and although there may be some tendency for the errors to average out (because households stagger their purchase of durable replacements to suit their incomes), there is no guarantee that this will be exact. In order to overcome this problem, the durable expenditures recorded in the HES were averaged across the different household types used in the comparisons, and the recorded durable expenditures on each item were replaced by the *average* expenditures on each durable good for each household type. The ABS itself argues that the use of group average expenditures in the case of items purchased infrequently will provide accurate estimates if the groups themselves are sufficiently large (ABS, 1995b).

Before comparing the budget standards in Table 1 with the HES data adjusted as described above, it is useful to describe some of the features of the (adjusted) HES data. Table 2 shows, for each of the household types that broadly correspond to the BSU categories (as defined in Table 1), the percentage of each household type falling below one-half of the *overall* median, between one-half of the median and the median itself and above the median. The estimates shown are based on total household expenditure and household disposable income, with the median in each case being derived after applying an equivalence adjustment equal to the square root of household size (to allow for variations in household need). (For information, the calculated values for adjusted (equivalised) median income and expenditure are \$317 and \$327, respectively. The corresponding unadjusted estimates are equal to \$509 and \$522).

The estimates in Table 2 show the extent to which the economic fortunes of different household types vary in Australia. The same general pattern is revealed by both the income-based and the expenditure-based figures, except for older households who appear much worse off, both absolutely and relative to others, when their standard of living is measured by their expenditure rather than their income - a somewhat surprising finding, though one confirmed in other research using the HES by Saunders (1997b). The proportion of the population with incomes or expenditures below one-half of the median is often taken as an indicator of relative low income or poverty. On this basis, Table 2 suggests that the half-median relative

	Adjusted Disposable Income ^(b)			Adjusted Expenditure ^(b)		
Household Type ^(a)	Below one-half of median	Between one-half of median and the median	Above the median	Below one-half of median	Between one-half of median and the median	Above the median
Single non-aged female	18.4	31.0	50.6	16.2	34.9	48.9
Non-aged couple	5.4	24.0	70.6	4.9	27.6	67.6
Couple plus one child	7.0	37.8	55.2	5.7	35.3	59.1
Couple plus two children	7.3	45.3	47.4	4.4	49.4	46.2
Couple plus three children	8.5	60.7	30.9	8.9	50.2	40.8
Couple plus four children	12.7	54.7	32.6	4.7	55.2	40.1
Sole parent plus one child	23.2	56.0	20.8	16.6	60.0	23.4
Sole parent plus two	27.8	59.3	12.9	24.2	59.6	16.2
children						
Aged single female	7.9	85.0	7.1	37.7	48.6	13.7
Aged couple	7.4	79.4	13.2	19.9	56.0	24.1
Other households	7.5	33.3	59.2	9.3	34.3	56.5
All households	8.2	41.8	50.0	11.5	38.6	50.0
Notes: a) Detailed characteristics of each household are shown in Table 1						

Table 2: The Estimated Distribution of H	lousehold Incomes and Expenditures by
Household Type: Updated to the March Q	Juarter 1997 (Percentages)

b) Recorded HES expenditure data for 1993-94 have been adjusted as described in the text and updated to the March Quarter 1997 by movements in the CPI.

1993-94 Household Expenditure Survey, confidentialised unit record file. Source:

poverty rate in 1997 was just over eight per cent measured on an income basis, or 11.5 per cent when measured using expenditure. The former figure is quite close to the 10.2 per cent poverty rate estimated using a half-median income poverty line by ABS for 1995-96 using data from the income distribution survey for that year (ABS, 1998:125), even though the ABS used a different equivalence adjustment.

On the basis of the expenditure figures, Table 2 implies that around 11 per cent of households spend less than one-half of median expenditure and thus could be considered to be living in relative 'expenditure poverty' (Saunders, 1997b). This percentage varies for different household types, ranging from less than five per cent for couples without children and some couples with children, to 24 per cent for sole parents with two children and almost 38 per cent for single aged women. These differences in the percentages falling

below the half-median benchmark are mirrored by the percentages lying above the median. Thus, while over two-thirds of couples without children have above-median expenditure, only around 20 per cent of sole parent and older households spend more than the overall median. Overall, the results in Table 2 present an illuminating picture of variations in the level and distribution of the living standards of different families, as measured by their incomes and expenditures relative to their needs.

We now compare actual living standards as measured by the HES data, adjusted as described earlier, with the BSU budget standard adequacy benchmarks presented in Table 1. Three sets of such comparisons are shown in Table 3. The first set of comparisons, shown on the left, restrict the coverage of the HES data to households living in (unfurnished) private rental accommodation, as this is the housing tenure to which the budget standards apply (except for older households where, for consistency, the HES data also refer to outright owners). These estimates are based on a rather small sub-sample, equivalent to just over 38 per cent of the full range of housing tenures for the identified household types shown in Table 2. Furthermore, the incomes of private renter households are generally below those living in other forms of housing tenure, which can serve to distort the impression conveyed by the results.

Because of this latter possibility, the second set of results in Table 3 includes, in addition to private renter households, owners, purchasers and those living in (unfurnished) public rental housing. The adjusted expenditures of these households are still compared with the budget standards of private renter households, partly because the scope of the BSU project means that comparisons are not always possible for other tenures, and even where they are, the BSU housing cost estimates (e.g. for purchasers) are not representative of the circumstances of all households. (It is also worth noting here that the budget standards do not vary greatly with housing tenure, particularly if housing costs themselves are excluded, so that the use of budgets for a single form of housing tenure will not greatly influence the general pattern of results).

	Privat househ	te renter olds only	Owners,	, purchasers a	nd renter households			
	(Inclu housin	usive of ng costs)	ive of (Including (E s costs) housing costs) hou		(Exc housir	ccluding ing costs)		
Household Type ^(b)	Below low cost standard	Below modest but adequate standard	Below low cost standard	Below modest but adequate standard	Below low cost standard	Below modest but adequate standard		
	20.4	50 7	26.2	5 4 <i>c</i>	25.0	40.0		
Single non-aged female	28.4	53.7	36.3	54.6	25.9	48.8		
Non-aged couple	11.2	28.6	10.4	27.6	4.8	17.7		
Couple plus one child	31.9	63.7	16.7	45.3	10.3	34.6		
Couple plus two children	47.1	82.2	30.4	64.9	12.5	50.7		
Couple plus three children	50.7	76.9	36.9	76.7	22.4	69.8		
Couple plus four children	43.6	91.1	32.9	85.4	22.8	69.7		
Sole parent plus one	46.7	80.4	56.4	83.0	28.0	77.4		
Sole parent plus two children	65.2	94.3	56.5	88.3	41.5	71.6		
Aged single female	56.5	79.7	57.3	80.3	56.2	78.0		
Aged couple	27.2	55.8	27.2	56.0	23.3	53.4		
All households	36.1	62.3	28.7	54.4	20.4	46.2		

Table 3: Percentages of Actual Household Expenditures Below BSU Budget Standard Adequacy Benchmarks for Low Cost and Modest but Adequate **Standards: February 1997**^(a)

the text and updated to the March Quarter 1997 by movements in the CPI.

b) Detailed characteristics of each household type are shown in Table 1.

Source: 1993-94 Household Expenditure Survey, confidentialised unit record file.

The final set of comparisons shown in Table 3 modify the second set by excluding housing costs from the HES expenditure data and the housing budget component of the budget standards. This variant not only addresses the unrepresentative nature of the BSU housing budgets for some tenure types, but also avoids the problems associated with the fact that the BSU housing budgets as a whole (which are relevant to the Hurstville LGA in Sydney) are unrepresentative of housing costs in other parts of the country (remembering that the HES data is derived from a *nationally* representative sample). In the light of the lack of correspondence between the HES and BSU data, and given the data limitations discussed above, this third set of comparisons, while limited in scope, is likely to be the most reliable.

What do the comparisons in Table 3 reveal? On an overall basis, the initial set of estimates indicates that over 36 per cent of private renter households were living below the BSU low cost standard in February 1997, with a further 26 per cent lying between that and the modest but adequate standard. These figures both decline by around eight percentage points when other housing tenure groups are included, and by a further eight percentage points when housing costs are excluded for the broader sample. Thus, the final set of estimates implies that just over 20 per cent of Australian households were unable to maintain a low cost standard net of housing costs in 1997, while around 46 per cent were living below a modest but adequate standard. Both of these latter figures are higher than the corresponding half-median and median expenditure-based comparisons shown in Table 2, although at 20.4 per cent, the percentage below the low cost standard (after excluding housing costs) is virtually identical to the estimated (before housing costs) income poverty rate in 1993-94 of 20.3 per cent derived from the HES data using the Henderson poverty line (Saunders, 1997b, Table 1) and the corresponding ABS estimated Henderson poverty rate of 20.5 per cent in 1995-96 (ABS, 1998: 125).

Within household types, the final set of estimates in Table 3 suggest that the worst off groups, with the greatest incidence of a less than low cost standard of living, are single older women, sole parents with one or two children and single non-aged women. These are the groups that have been identified in conventional poverty research as experiencing the greatest risk of poverty. If affluence is assessed in terms of having the largest percentage above the modest but adequate standard, then Table 3 indicates that, on average, the most affluent households are couples without children, followed by couples with one or two children and older couples. One of the most marked features of all three sets of results in Table 3 that warrants further investigation is the considerable difference between the implied standard of living of older single women and older couples. In terms of absolute numbers, the final set of estimates in Table 3 imply that around 686 000 households containing 991 000 adults and just over 404 000 children were living below the low

cost standard in 1997. (These numbers appear on the low side because the analysis is restricted in its coverage to only just over half of all households included in the HES).

Overall, the third set of estimates in Table 3, in combination with the estimates shown on the right hand side of Table 2, confirm that the BSU low cost budget standards are somewhat higher than a level corresponding to one-half of the median standard in the community (as measured by household expenditure). Given that the one-half median benchmark is often used as a relative poverty standard, this implies that the low cost standard may be somewhat more generous than what is generally understood by poverty standard. Even so, the fact that around one-fifth of all households fall below the low cost standard after deducting housing costs, while the ABS (1998) estimates that the same proportion were below the Henderson poverty line (before housing costs) in 1995-96 suggests that the two measures are not that far apart once one removes the distorting effect of the BSU housing budgets.

4 The Sensitivity of the BSU Budget Standards

Several comments have already been made about the need to assess the sensitivity of a budget standard to changes in some of its key assumptions and judgments. Economists have argued the case for sensitivity analysis in areas such as poverty and income distribution in order to assess whether specific conclusions alter when particular value judgements are changed. Such analysis can help to establish how robust claims are that poverty or inequality have risen, to changes in where the poverty line is set or in how the incomes of different groups are combined when measuring inequality. The role and usefulness of sensitivity analysis in the context of the BSU budget standards are now illustrated with the half of two related examples: housing costs and spatial variations in prices.

Housing Costs

It has already been noted that the BSU housing budgets, which were derived for one LGA in Sydney are unlikely to be representative of housing costs in other parts of the country. This was confirmed in evidence provided to the BSU researchers by the focus groups, particularly those living in rural areas (Saunders et al., 1998, Chapter 13), and the variations shown in Table 3 provides further supporting evidence. The BSU housing budgets for private renter households were based on the levels of the median and first quartile rents in the Hurstville LGA for one, two and three bedroom units. The resulting low cost rents (in February 1997) were \$130, \$165 and \$212 a week, respectively. The corresponding rent levels incorporated into the modest but adequate standards were \$148, \$180 and \$230, respectively.

How do these rents compare with rents in other parts of Sydney and in other state capitals? One way to answer this question is to develop a statistical model which seeks to explain the variation in market rents by variables reflecting the quality and location of the dwelling and then to use the estimated model to predict the effect of differing locations on the rents of apartments with *given* characteristics. This approach allows for considerable flexibility but produces results that are only as good as the model on which they are based, although Henman (1998, forthcoming) has shown that the method is capable of providing a useful indication of how the BSU housing (and total) budgets vary with location. An alternative approach, developed below, avoids the use of an estimated model, relying instead on published data on the distribution of rents in different locations around Australia.

The source of these data is The Real Estate Institute of Australia (REIA) whose monthly review of major residential property markets in Australia, *Market Facts* which provides data on the distribution of the rents of three bedroom houses and two bedroom apartments which were *let in the previous month*. The latter REIA data (REIA, 1997) have been used to estimate the first quartile and median rents for different sized apartments in each capital city in February 1997, assuming that the ratio of rents for one and three bedroom units relative to rents for two bedroom units is the same elsewhere in Australia as in the Hurstville LGA. The resulting rent estimates are shown in Table 4.

	First	t Quartile R	lents	Ν	Median Rents			
Location	1	2	3	1	2	3		
	bedroom	bedroom	bedroom	bedroom	bedroom	bedroom		
	unit	unit	unit	unit	unit	unit		
			Dol	llars				
Hurstville LGA	130	165	212	148	180	230		
Sydney	147	186	239	173	210	268		
Melbourne	100	127	163	124	151	193		
Brisbane	110	139	179	129	157	200		
Adelaide	79	100	128	103	125	159		
Perth	71	90	116	90	110	140		
Canberra	102	129	166	119	145	185		
Hobart	82	104	134	97	118	150		
Note: a) All figures have been rounded to the nearest dollar.								

Table 4: Estimated First Quartile and Median Rents by Capital City: February1997 (Based on Units Let)^(a)

The first thing to note about Table 4 is that it indicates that rents in the Hurstville LGA are actually somewhat *below* those prevailing in Sydney as a whole. The difference in the case of two-bedroom units is around \$20 a week (or 13 per cent) at the low cost standard, and \$30 a week (or 17 per cent) at the modest but adequate standard. Even greater differences can be seen in capital cities other than Sydney, particularly those outside of the eastern seaboard.

Australia (1997).

Table 5 takes the analysis a step further by substituting the rental figures shown in Table 4 into budget standards for two of the household types shown in Table 1. For a couple with two children, the low cost standard estimated for the Hurstville LGA falls by as much as \$72 a week (12 per cent) for households living in Hobart or by \$88 a week (15 per cent) for those living in Perth. For sole parents with one child, the corresponding declines are \$56 a week (15 per cent) and \$69 a week (19 per cent), respectively. The magnitude of these variations raises the question of whether they would be accepted as the basis for varying the rates of social

	Couple with (girl, 6 a in 3 bec	Couple with two children (girl, 6 and boy, 14) in 3 bedroom unit		t with one child girl, 6) droom unit
Location	Low cost	Low Modest but cost adequate		Modest but adequate
Hurstville LG	A 602.1	817.4	371.8	519.8
Sydney	627.0	852.4	391.1	547.4
Melbourne	557.1	783.4	336.8	493.1
Brisbane	571.8	789.8	347.9	498.6
Adelaide	524.9	752.1	312.0	469.2
Perth	513.8	734.6	302.8	455.4
Canberra	559.8	776.0	338.7	487.6
Hobart	530.4	743.8	315.7	462.8
Sources:	Saunders et al., 1998, Australia (1997).	Tables 12.2 and	12.3 and Real	Estate Institute of

Table 5:	Illustrative	Effects o	of Loca	tion on	the	BSU	Budget	Standards	for	Private
Renter H	ouseholds: Fo	ebruary	1997 (1	Dollars	per w	veek)	_			

security support provided to households living in different parts of the country. If the answer to this question is in the negative, the case for criticising a budget standard for failing to recognise the same variations must itself come into question.

The figures in Table 5 illustrate the importance of housing costs in the budget standards, but they should not be accepted uncritically. It is important to recognise, for example, that they provide an incomplete and hence imperfect estimate of the effect of changing location on a budget standard. Of significance here is the fact that changing the location of the household will affect parts of its budget standard other than just housing costs, because several other aspects of the BSU budgets are specific to the assumption that the household lives in the selected location. For example, changes in location will impact upon the household's transport budget which is based on the distances travelled to shops, schools, places of work, child care centres and other services in a specific locality. More significantly, a new location may mean that the household cannot take advantage of the competitive prices available in major shopping centres,

with the result that the entire budget has to be repriced. Movements across State borders are likely to affect the range of State Government services available or how they are subsidised, with obvious implications for a budget standard.

There is also the fact that a different location may change the standard of living that corresponds to a particular type of dwelling, implying that the entire budget may have to be redeveloped. The more that the significance of these kinds of conceptual and practical issues is acknowledged, the clearer it becomes that estimates like those shown in Table 5 are likely to be subject to considerable margins of error. The bottom line of this discussion is that there is no substitute for respecifying and repricing the entire budget for each new location - an exercise that, whilst time-consuming and expensive, is the only way to produce authoritative estimates of the sensitivity of a budget standard to its location.

Spatial Price Variations

The discussion to date has revealed the crucial role played by the pricing structure in determining what it costs to maintain a specific standard of living. Different prices were used in the BSU strategy to distinguish between the low cost and modest but adequate standards and, as the earlier discussion highlights, are a very important component of the housing budget. However, the prices used to do this were those prevailing in Sydney (often those in the Hurstville LGA) and the question that now arises is how one might adjust the Sydney standards so that they more accurately reflect costs in other locations. The housing costs issue just discussed is one aspect of this, but the more general issue of price variations in different locations is now addressed.

Although the discussion focuses on *spatial* variations in prices, it is worth observing at the outset that there are other important aspects of pricing that also warrant further investigation. One of these concerns the adjustment of a budget standard over time as the average price level changes. This issue was discussed in the final chapter of the budget standards report (Saunders et al., 1988), where it was noted that since the BSU budgets apply to Sydney, for

consistency they should also be adjusted in line with movements in Sydney prices (at least in the short to medium term). The other aspect of price variations relates to changes in the relative prices of different items in the budgets, as will occur, for example, with the introduction of a GST. This issue is currently being explored within the BSU and will form the basis of a report due for release later in the year.

Despite the interest in the topic, both among researchers and policy analysts, there is relatively little systematic and comprehensive information on the spatial variation in consumer prices. The topic has been studied by EPAC (Ferry, 1991) and King (1995), both of whom have reviewed what evidence is available. Ferry presents estimates of relative costs of living in seven capital cities and shows that they display considerable variation, not only between cities, but also over time and according to housing tenure (Ferry, 1991, Tables 2 to 4). He shows that Sydney prices are consistently well above those in all other cities for renters and home buyers, though not for home owners.

King (1995) presents evidence collected by the Australian Consumers Association on the cost of a grocery basket of brand name products in 23 cities in 1993. The estimates range from around \$62 a week to \$78, with 17 of the 23 cities falling in the relatively narrow range between \$65 and \$70 (King, 1995: 59). Interestingly, Sydney falls within this range, but towards the *lower* end, with a basket priced at \$67 (King, 1995, Figure 2). On balance, King concludes that:

... while the evidence on regional variations in the cost of living in Australia is patchy, the available evidence does strongly suggest the possible existence of significant variations in costs other than the well known variation in housing costs. (King, 1995: 66)

On this basis, it would therefore seem that the entire BSU budget standards may need to be repriced if they are to accurately reflect the circumstances of households living outside of Sydney. The evidence reviewed by Ferry (1991) and King (1995) does not carry the authority of the Consumer Price Index (CPI) estimated and published by ABS. The issue of whether ABS itself should develop regional cost of living estimates (as opposed to estimates of the *movements* in consumer prices in different capital cities, which it currently does provide), has been considered as part of the recent review of the CPI, but was not identified as a priority for future development (ABS, 1997a). The main reason for this appears to lie in the difficulty of estimating appropriate weights at the regional or city level which would be representative of the expenditure patterns of households in the area. Without such weights, it is not possible to derive a reliable index of consumer prices for each region which can form the basis of price and cost of living comparisons.

In spite of this, ABS has been collecting and publishing lists of the average retail prices of selected grocery items in each of the eight capital cities (ABS, 1997b). The list includes the prices of a variety of foodstuffs (including diary products, meat, fresh fruit and vegetables and processed foods), household supplies, petrol, alcoholic drinks and personal care items. Figure 1 shows, for the March Quarter 1997 (when the BSU budgets were costed) the variations in selected items compared to Sydney and the variation in the total cost of all 57 items covered in the ABS publication. Although it is clear that some of the variations cancel out others, the fact that most of the variations in Figure 1 are negative confirms the fact that prices in Sydney are above those in most other cities. Of course, these comparisons suffer from the absence of a method for weighting the different items: simply adding up the price of all 57 items as is done in the final panel of Figure 1 assigns an equal weight to each item, which is clearly not an appropriate representation of actual spending patterns.

Further evidence on grocery expenditure variations is provided in a recent study conducted by A.C. Nielsen for the Australian Supermarket Institute (ASI, 1998). Participants surveyed as part of the study were asked: 'On average, approximately how much would you say you spend on all the household's food and grocery per week?'. The highest mean response

Figure 1: Price Differences for Selected Commodities: March Quarter 1997 (Percentage difference from Sydney prices)



Melbourne Brisbane Adelaide Perth Hobart Darwin Canberra 25 1 kg laundry detergent 20 15 10 5 0 -5 -10 -15 -20 -25 25 1 litre unleaded petrol 20 15 10 5 0 -5 -10 -15 -20 -25 1 25 2 x 125g toilet soap 20 15 10 5 0 -5 -10 -15 -20 -25 🕹 25 All 57 items 15 5 -5 -15 -25 L

Figure 1: Price Differences for Selected Commodities: March Quarter 1997 (Percentage difference from Sydney prices) (Continued)

(\$137) was provided by those living in Sydney. This was 8.7 per cent above the national average, with the averages for other cities falling below the Sydney figure by 5.1 per cent (Melbourne), 10.2 per cent (Brisbane), 28.5 per cent (Adelaide), 16.1 per cent (Perth) and 18.2 per cent (Tasmania). These figures refer to variations in *expenditure* and thus incorporate differences in both the prices paid and the quantities consumed and (because Sydney households tend to have the highest average incomes and thus consume more) will overstate the differences in prices alone. They are also based on a smaller sample than the ABS data summaries in Figure 1 and are thus somewhat less reliable. Even so, the available comparisons provide further support to King's (1995) conclusion that there is evidence that prices vary systematically by location. It is also worth observing that Sydney appears to have *both* the highest housing costs and the highest grocery costs. There is no general tendency for relatively low housing costs to be cancelled out by relatively high grocery prices, although Table 4 and the ASI figures quoted above suggest that this does occur to some extent in places like Adelaide, Perth and Tasmania.

In summary, the available evidence on spatial price variations suggests that it would be necessary to reprice the Sydney-based BSU budget standards in each capital city (and possibly in regional centres also) before they can be used to provide a basis for assessing the adequacy of the incomes of households living in different parts of Australia. This is a big ask and not one that is currently capable of being addressed by any other living standards indicator. Once again, it is important not to be too critical of what a budget standard *cannot* do in achieving what no other measure has yet succeeded in doing. Having said this, however, it is clear that there is considerable scope for further research on the spatial aspects of budget standards in order to improve their accuracy and relevance.

5 Conclusion

This paper has described how the BSU budget standards were developed, discussed some of their strengths and weaknesses, and illustrated what role budget standards can play in assessing household living standards and the

adequacy of household incomes. The single most important contribution that budget standards research can make to these issues is in providing a coherent and transparent framework for thinking about the items needed to maintain a particular material standard of living and translating them through prices into the budgets required to purchase them.

The discussion in this paper has been motivated by the view that budget standards have a very important role to play in the living standards debate in Australia. Having said that, there is considerable room for improving the existing budget standards, possibly by changing some of the judgements that underlie them, but also by repricing them in cities and regions other than Sydney. The results presented in Sections 3 and 4 of the paper suggest that this has the potential to greatly influence what conclusions can be drawn about the adequacy of household incomes in relation to the low cost and modest but adequate standards. The budget standards research already undertaken is an important step forward, but there is much more work to be done and many more challenges to be faced along the way.

References

- Australian Bureau of Statistics (ABS) (1995a), 1993-94 Household Expenditure Survey. Australia. User Guide, Catalogue No. 6527.0, ABS, Canberra.
- ABS (1995b), 1993-94 Household Expenditure Survey, Australia. Summary of Results, Catalogue No. 6530.0, ABS, Canberra.
- ABS (1997a), Information Paper. Outcome of the 13th Series Australian Consumer Price Index Review, Catalogue No. 6453.0, ABS, Canberra.
- ABS (1997b), Average Retail Prices of Selected Items. Eight Capital Cities. March Quarter 1997, Catalogue No. 6403.0, ABS, Canberra.
- ABS (1998), Australian Social Trends. 1998, Catalogue No. 4102.0, ABS, Canberra.
- Australian Supermarket Institute (ASI) (1998), *The Way We Shop. Grocery Shopping in Australia*, ASI, Sydney.
- Brownlee, H. (1990), *Measuring Living Standards*, Paper No. 1, Australian Living Standards Study, Australian Institute of Family Studies, Melbourne.
- Ferry, N. (1991), 'Regional income levels and living standards in Australia', in *Background Papers on Urban and Regional Issues*, Background Paper No. 10, Economic Planning Advisory Council, Canberra, 61-101.
- Henman, P. (1998), 'Different assumptions, different scenarios: extending and developing SPRC's Budget Standards', *Social Security Journal*, forthcoming.
- King, A. (1995), 'The case for a regional dimension in income support', in P. Saunders, ed., Social Policy and Northern Australia: National Policies and Local Issues, SPRC Reports and Proceedings No. 120, Social Policy Research Centre, University of New South Wales, Sydney, 55-74.
- Real Estate Institute of Australia (REIA) (1997), Market Facts. February 1997, REIA, Canberra.
- Renwick, T.J. (1993), 'Budget-based poverty measurement: 1992 Basic Needs Budgets', Proceedings of the American Statistical Association, Social Statistics Section, 573-82.

- Renwick, T.J. and B.R. Bergmann (1993), 'A budget-based definition of poverty with an application to single-parent families', *Journal of Human Resources*, XXV111(1), 1-24.
- Ruggles, P. (1990), Drawing the Line: Alternative Poverty Measures and Their Implications for Public Policy, The Urban Institute Press, Washington D.C.
- Saunders, P. (1997a), 'Economic adjustment and distributional change: income inequality and poverty in Australia in the 1980s', in P. Gottschalk, B. Gustafsson and E. Palmer, eds, *Changing Patterns in the Distribution of Economic Welfare. An International Perspective,* Cambridge University Press, Cambridge, 60-83.
- Saunders, P. (1997b), 'Living standards, choice and poverty', Australian Journal of Labour Economics, 1(1), 49-70.
- Saunders, P., J. Chalmers, M. McHugh, C. Murray, M. Bittman and B. Bradbury (1998), *Development of Indicative Budget Standards for Australia*, Policy Research Paper No. 74, Department of Social Security, Canberra.
- Sen, A. K. (1998), *The Standard of Living*, Cambridge, Cambridge University Press.
- Social Welfare Policy Secretariat (1981), Report on Poverty Measurement, AGPS, Canberra.
- Travers, P. and S. Richardson (1993), *Living Decently. Material Well-Being in Australia*, Oxford University Press, Melbourne.
- Watts, H.W. (1980), New American Budget Standards: Report of the Expert Committee on Family Budget Revisions, Special Report Series, Institute for Research on Poverty, University of Wisconsin, Madison.

SOCIAL POLICY RESEARCH CENTRE DISCUSSION PAPERS

- No longer available.
- ♦ Published in Journal (list follows)

1. * 令	The Labour Market Position of Aboriginal People in Non-Metropolitan New South Wales	Russell Ross	August 1988
2. *	Welfare Fraud, Work Incentives and Income Support for the Unemployed	Bruce Bradbury	August 1988
3.◆◇	Taxation and Social Security: An Overview	Peter Whiteford	August 1988
4. ◆ ◇	Income Inequality in Australia in an International Comparative Perspective	Peter Saunders & Garry Hobbes	August 1988
5. ◆ 令	Family Size Equivalence Scales and Survey Evaluations of Income and Well-Being	Bruce Bradbury	December 1988
6. * 令	Income Testing the Tax Threshold	Peter Whiteford	December 1988
7. *	Workers' Compensation and Social Security Expenditure in Australia: Anti-Social Aspects of the 'Social' Wage	Don Stewart & Jennifer Doyle	December 1988
8. *	Teenagers in the Labour Market: 1983-1988	Russell Ross	December 1988
9. *	A Legacy of Choice: Economic Thought and Social Policy in Australia, the Early Post-War Years	Paul Smyth	May 1989
10. * 令	The 'Family Package' and the Cost of Children	Bruce Bradbury	May 1989
11.*	Towards an Understanding of Commonwealth Social Expenditure Trends	Peter Saunders	May 1989
12. ◆ 令	A Comparative Study of Home and Hospital Births: Scientific and Normative Variables and their Effects	Cathy Boland	July 1989
13.*	Adult Goods and the Cost of Children in Australia	Bruce Bradbury	July 1989
14. * 令	Some Australian Evidence on the Consensual Approach to Poverty Measurement	Peter Saunders & Bruce Bradbury	July 1989
15 🗇	Income Inequality in Australia and New Zealand: International Comparisons and Recent Trends	Peter Saunders, Garry Hobbes & Helen Stott	September 1989
16. * 令	Trends in the Disposable Incomes of Australian Families, 1982-83 to 1989-90	Bruce Bradbury, Jennifer Doyle & Peter Whiteford	January 1990

17.\$	Selectivity and Targeting in Income Support: The Australian Experience	Peter Saunders	February 1990
18. * 令	How Reliable are Estimates of Poverty in Australia? Some Sensitivity Tests for the Period 1981-82 to 1985-86	Bruce Bradbury & Peter Saunders	February 1990
19. 🍫	The Labour Supply Behaviour of Single Mothers and Married Mothers in Australia	Russell Ross & Peter Saunders	July 1990
20. * \$	Income Poverty Among Aboriginal Families with Children: Estimates from the 1986 Census	Russell Ross & Peter Whiteford	July 1990
21.\$	Compensating Low Income Groups for Indirect Tax Reforms	Peter Saunders & Peter Whiteford	August 1990
22. * 令	Reflections on the Review of the Home and Community Care Program	Peter Saunders	August 1990
23. ◆ 令	Sole Parent Families in Australia	Peter Saunders & George Matheson	September 1990
24.\$	Unemployment, Participation and Family Incomes in the 1980s	Bruce Bradbury	September 1990
25. ◆ 令	Employment Growth and Poverty: An Analysis of Australian Experience, 1983-1990	Peter Saunders	September 1990
26. *	Gender, Social Policy Regimes and the Welfare State	Sheila Shaver	November 1990
27.	A Probit Analysis of the Factors Influencing Labour Market Success of Aborigines in New South Wales	Russell Ross	November 1990
28. ◆ 令	Efficiency and Effectiveness in Social Policies: An International Perspective	Peter Saunders	December 1990
29.	Take-up of Family Income Supplement in 1986 - A Research Note	Peter Whiteford & Jennifer Doyle	February 1991
30.\$	An Ever-Rising Tide? Poverty in Australia in the Eighties:	Peter Saunders & George Matheson	May 1991
31.\$	Are Immigrants Over-Represented in the Australian Social Security System?	Peter Whiteford	March 1992
32.	Measuring the Cost of Children	Bruce Bradbury	May 1992
33.\$	The Impact of Family Assistance Changes on Patterns of Unemployment Benefit Receipt	Bruce Bradbury	August 1992
34.\$	Recent Trends in the Size and Growth of Government in OECD Countries	Peter Saunders	September 1992

35.\$	Noncash Income, Living Standards, Inequality and Poverty: Evidence from the Luxembourg Income Study	Peter Saunders et al	November 1992
36. ◆ 令	The Mixed Economy of Support for the Aged In Australia: Lesson for Privatisation	Peter Saunders & Michael Fine	November 1992
37.	The Welfare Interpretation of Family Size Equivalence Scales	Bruce Bradbury	November 1992
38.\$	Body Rights, Social Rights and the Liberal Welfare State	Sheila Shaver	December 1992
39.\$	Unemployment and Income Support: Challenges for the Years Ahead	Bruce Bradbury	May 1993
40.\$	Married Women's Earnings and Family Income Inequality in the Eighties	Peter Saunders	May 1993
41.	Women and the Australian Social Security System: From Difference Towards Equality	Sheila Shaver	June 1993
42.	Male Wage Inequality Before and After Tax: A Six Country Comparison	Bruce Bradbury	June 1993
43. *	The Fragmented Structure of Community Support Services: A Community Case Study	Michael Fine	June 1993
44. ◆ ◇	The Recognition of Wifely Labour by Welfare States	Sheila Shaver & Jonathan Bradshaw	August 1993
45.	Postmodernism and Social Policy: A Great Leap Backwards?	Peter Taylor-Gooby	September 1993
46. ^{\$}	Making Ends Meet in Australia and Sweden: A Comparative Analysis of the Consensual Approach to Poverty Measurement	Peter Saunders, Björn Halleröd & George Matheson	October 1993
47. \$	Economic Adjustment and Distributional Change: Income Inequality and Poverty in Australia in the Eighties	Peter Saunders	November 1993
48. \$	Poverty and Inequality: Social Security in Australia in the 1990s	Peter Saunders	May 1994
49.\$	Rising on the Tasman Tide: Income Inequality in Australia and New Zealand in the 1980s	Peter Saunders	June 1994
50.	A New Approach to the Direct Measurement of Consensual Poverty	Björn Halleröd	October 1994
51.	The Distribution of Welfare: Inequality, Earnings Capacity and Household Production in a Comparative Perspective	Peter Saunders Inge O'Connor & Timothy Smeeding	November 1994

52.\$	Immigrants and the Distribution of Income: National and International Comparisons	Peter Saunders	November 1994
53.\$	The Role, Value and Limitations of Poverty Research	Peter Saunders	November 1994
54.\$	The Use of Replacement Rates In International Comparisons of Benefit Systems	Peter Whiteford	February 1995
55.\$	Two Papers on Citizenship and the Basic Income	Peter Saunders & Sheila Shaver	April 1995
56.\$	Improving Work Incentives in a Means-tested System: The 1994 Australian Social Security Reforms	Peter Saunders	May 1995
57. \$	Corporatism in Australia	Peter Kriesler & Joseph Halevi	May 1995
58.	Universality and Selectivity in Income Support: A Comparative Study in Social Citizenship	Sheila Shaver	May 1995
59.	Household Semi-public Goods and the Estimation of Consumer Equivalence Scales: Some First Steps	Bruce Bradbury	May 1995
60.\$	Wage and Income Inequality in Two Welfare States: Australia and Sweden	Peter Saunders & Johann Fritzell	August 1995
61. *	The Changing Mix of Welfare in Health Care and Community Support Services	Michael Fine	August 1995
62.\$	Evaluation and Research in Social Policy	Peter Saunders I & Michael Fine	December 1995
63.\$	Unpacking Inequality: Wage Incomes, Disposable Incomes and Living Standards	Peter Saunders I	December 1995
64. ◆ 令	A Challenge to Work and Welfare: Poverty in Australia in the 1990s	Peter Saunders I	December 1995
65. ◆ 令	Social Policy and Personal Life: Changes	Sheila Shaver & l	December
	in State, Family and Community in the Support of Informal Care	Michael Fine	
66.	Household Income Sharing, Joint Consumption and the Expenditure Patterns of Australian Couples and Single People	Bruce Bradbury	May 1996
67.	Explaining Changes in the Social Structure of Employment: The Importance of Geography	Boyd Hunter	June 1996
68.	Liberalism, Gender and Social Policy	Sheila Shaver	July 1996
69.	Redistribution by the State in Austria	Alois Guger	October 1996

70.	Economic Crisis and Social Policy in Finland in the 1990s	Hannu Uusitalo	October 1996
71.	Sole Mothers in Australia: Supporting Mothers	Marilyn McHugh	November
	to Seek Work	& Jane Millar	
72.	'All Else Confusion': What Time Use Surveys	Michael Bittman	November
	Show About Changes in Gender Equity	& George Mathes	on
73.	Are the Low Income Self-employed Poor?	Bruce Bradbury	December 1996
74.	Social Policy in East Asia and the Pacific Area in the Twenty-First Century: Challenges and Responses	Peter Saunders	December 1996
75.	Dawning of a New Age? The Extent, Causes and Consequences of Ageing in Australia	Peter Saunders	December 1996
76.	Poverty, Choice and Legitimacy	Peter Saunders	March 1997
77.	The Restructuring of the Canadian Welfare State: Ideology and Policy	Maureen Baker	June 1997
78.	Developing Policy Planning and Research Capabilities in the Asia Pacific	Peter Saunders	October 1997
79.	New Relations of Welfare in the Contracting State: The Marketisation of Services for the Unemployed in Australia	Tony Eardley	October 1997
80.	Coordinating Health, Extended Care and Community Support Services: Issues for Policy Makers and Service Providers in Australia	Michael Fine	October 1997
81.	How do the Elderly in Taiwan Fare Cross- Nationally? Evidence from the Luxembourg Income Study Project	Peter Saunders & Timothy M. Smeeding	April 1998
82.	An Australian Model for Labour Supply and Welfare Participation in Two-adult Households	Guyonne Kalb	June 1998
83.	The Land of the Lost Long Weekend? Trends in Free Time Among Working Age Australians, 1974-1992	Michael Bittman	June 1998
84.	Defining Poverty and Identifying the Poor: Reflections on the Australian Experience	Peter Saunders	June 1998

85.	An Equivalence Scale for Time	Michael Bittman & Robert E. Goodin	July 1998
86.	The Changing Boundary Between Home and Market: Australian Trends in Outsourcing Domestic Labour	Michael Bittman, Gabrielle Meagher & George Matheso	July 1998 on
87.	Incomes, Incentives and the Growth of Means Testing in Hungary	Gerry Redmond	August 1998
88.	Economic Insecurity	Lars Osberg	October 1998
89.	Household Budgets and Income Distribution Over the Longer Term: Evidence for Australia	Peter Saunders	October 1998
90.	Global Pressures, National Responses: The Australian Welfare State in Context	Peter Saunders	October 1998
91.	Working But Poor? Low Pay and Poverty in Australia	Tony Eardley	November 1998
92.	Extension Amidst Retrenchment: Gender and Welfare State Restructuring in Australia and Sweden	Sheila Shaver	December 1998
93.	Using Budget Standards to Assess the Well- Being of Families	Peter Saunders	December 1998
94.	Later Life, Gender and Ethnicity: Changing Theory for Social Policy Research	Gail Wilson	December 1998
95.	Social Participation and Family Welfare: The Money and Time Costs of Leisure	Michael Bittman	February 1999
96.	The Increasing Financial Dependency of Young People on Their Families	Judy Schneider	February 1999
97.	The Rush Hour: The Quality of Leisure Time and Gender Equity	Michael Bittman & Judy Wajcman	February 1999
98.	Women and Retirement Income in Australia: Social Rights, Industrial Rights and Property Rights	Merrin Thompson	May 1999
99.	The 'Dutch Miracle': Employment Growth in a Retrenched but Still Generous Welfare System	Uwe Becker	May 1999
100.	Tax Theory and Targeting: A Survey	Bruce Bradbury	May 1999
101.	Home and Away: Reflections on Long-term Care in the UK and Australia	Melanie Henwood	June 1999

102.	Australian Attitudes to Unemployment and Unemployed People	Tony Eardley and George Matheson	June 1999
103.	The Costs of Children: Budget Standards Estimates and the Child Support Scheme	Marilyn McHugh	July 1999
104.	Tax-benefit Policies and Parents' Incentives to Work: The Case of Australia 1980-1997	Gerry Redmond	July 1999
105.	The Responsibility for Child and Aged Care: Shaping Policies for the Future	Michael Fine	August 1999

SOCIAL POLICY RESEARCH CENTRE REPRINTS

The following Discussion Papers have been published in journals or books. Where indicated, Reprints of the articles are available from the SPRC at the cost of \$2.00 each. To order reprints, quote the Reprint number and attach a cheque or money order made out to the Social Policy Research Centre. Send orders to:

The Publications Officer Social Policy Research Centre University of New South Wales Sydney NSW 2052 Australia

DP No.	Published as	SPRC Reprint No. if applicable)
1.	Russell Ross (1988), 'The Labour Market Position of Aboriginal People in Non-metropolitan New South Wales', <i>Australian</i> <i>Bulletin of Labour</i> , 15(1), December, 29-56.	48
3.	Peter Whiteford (1989), 'Taxation and Social Security: An Overview', <i>Australian Tax Forum</i> , 6(1), 2-39.	49
4.	Peter Saunders and Garry Hobbes (1988), 'Income Inequality in an International Comparative Perspective,' <i>Australian</i> <i>Economic Review</i> , 3rd Quarter, 25-34.	47
5.	Bruce Bradbury (1989), 'Family Size Equivalence Scales and Surv Evaluations of Income and Well-being', <i>Journal of Social Policy</i> , 18(3), July, 383-408.	ey 52
6.	Peter Whiteford (1989), 'Taxation Reform and the Tax Threshold' in John G. Head, ed., <i>Australian Tax Reform in Retrospect and</i> <i>Prospect</i> , papers presented at a conference organised by the Centre of Policy Studies, Monash University, Conferences Series no. 8, Australian Tax Research Foundation, Sydney, 219-47.	2
10.	Bruce Bradbury (1989), 'The "Family Package" and the Cost of Children', <i>Australian Social Policy</i> , 1(12), Winter, 21-51.	59

DP No.	Published as	SPRC Reprint No. (if applicable)
12.	Cathy Boland (1989), 'A Comparative Study of Home and Hospit Births: Scientific and Normative Variables and Their Effects', in <i>Celebrating a Revolution in Birth</i> : Proceedings of 10th National Homebirth Conference, Sydney, 19-33.	al
14.	Peter Saunders and Bruce Bradbury (1991), 'Some Australian Evidence on the Consensual Approach to Poverty Measurement', <i>Economic Analysis and Policy</i> , 21(1), March, 47-73.	62
15.	Peter Saunders, Helen Stott and Garry Hobbes (1991), 'Income Inequality in Australian and New Zealand: International Comparisons and Recent Trends', <i>Review of Income and Wealth</i> , 37(1), March, 63-79.	47
16.	Bruce Bradbury, Jenny Doyle and Peter Whiteford (1993), 'Trend in the Disposable Income and Housing Costs of Australian Famili Greg Mahoney, ed., <i>The Australian Economy under Labor</i> , Allen and Unwin, Sydney, 137-158.	s es', 71
17.	Peter Saunders (1991), 'Selectivity and Targeting in Income Support: The Australian Experience', <i>Journal of Social Policy</i> , 20(3), 299-326.	
18.	Bruce Bradbury and Peter Saunders (1990), 'How Reliable are Estimates of Poverty in Australia? Some Sensitivity Tests for the Period 1981-82 to 1985-86', <i>Australian Economic Papers</i> , 29(55), December 154-81.	60
19.	Russell Ross and Peter Saunders (1993), 'The Labour Supply of Sole Mothers and Married Mothers in Australia: Evidence from the 1986 Income Distribution Survey', <i>Australian Economic Pape</i> Vol. 32, June, 116-133.	ers,
20.	Russell Ross and Peter Whiteford (1992), 'Poverty in 1986: Aboriginal Families with Children', <i>Australian Journal of Social Issues</i> , 27(2), May, 92-111.	61
21.	Peter Saunders and Peter Whiteford (1990), 'Compensating Low Income Groups for Indirect Taxes', <i>Australian Tax Forum</i> , 7(4), 443-64.	
22.	Peter Saunders (1990), 'Reflections on the Review of the HACC Program', in A. Howe, E. Ozanne and C. Selby Smith, eds, <i>Community Care Policy and Practice: New Directions in</i> <i>Australia</i> , Public Sector Management Institute, Monash University, Victoria, 201-12.	63
23.	Peter Saunders and George Matheson (1991), 'Sole Parent Families in Australia', <i>International Social Security Review</i> , 44(3), 51-75.	

DP No. Published as

SPRC Reprint No. (if applicable)

- 24. Bruce Bradbury (1992), 'Unemployment, Participation and Family Incomes in the 1980s', *Economic Record*, 68(203), December, 328-42.
- 25. Peter Saunders (1991), 'Employment Growth and Poverty: An Analysis of the Australian Experience 1982-1990', in Michael Johnson, Peter Kriesler and Anthony D. Owen, eds, *Contemporary Issues in Australian Economics*, The Economic Society of Australia, Macmillan, Australia, 105-33. (Also excerpts in *ACTCOSS News*, 5 October, 12-14.)
- 28. Peter Saunders (1991), 'Efficiency and Effectiveness in Social Policies: an International Perspective', in T. P. Hardiman and Michael Mulreany, eds, *Efficiency and Effectiveness in the Public Domain*, Institute of Public Administration, Dublin, 78-117.
- 30. Peter Saunders and George Matheson (1991), 'An Ever Rising Tide?: Poverty in Australia in the Eighties', *Economic and Labour Relations Review*, 2(2), December, 142-71.
- 31. Peter Whiteford (1991), 'Are immigrants over-represented in the Australian social security system?', *Journal of the Australian Population Association*, 8(2), November, 93-109.
- Bruce Bradbury (1993), 'Family Assistance and the Incomes of Low Wage Families', *Social Security Journal*, March, 1-18. and
 Bruce Bradbury (1993), 'Family Assistance, Replacement Rates and the Unemployment of Married Men', *Australian Bulletin of Labour*, Vol. 19, No. 2, June, 114-132.
- 34. Peter Saunders (1993), 'Recent Trends in the Size and Growth of Government in OECD Countries', in Normal Gemmell, ed., *The Growth of the Public Sector: Theories and International Evidence*, Edward Elgar, Aldershot, 17-33.
- 35. Timothy M. Smeeding, Peter Saunders, John Coder, Stephen Jenkins, Johan Fritzell, Aldi J. M. Hagenaars, Richard Hauser and Michael Wolfson (1993), 'Poverty, Inequality and Family Living Standards Impacts Across Seven Nations: The Effects of Noncash Subsidies for Health, Education and Housing', *The Review of Income and Wealth*, Series 39, No. 3, September, 229-256.
- 36. Peter Saunders and Michael Fine (1992), 'The Mixed Economy of Support for the Aged in Australia: Lessons for Privatisation', *Economic and Labour Relations Review*, 3(2), December, 18-42.
- 38. Sheila Shaver (1993), 'Body Rights, Social Rights and the Liberal Welfare State', *Critical Social Policy*, Issue 39, Winter 1993/94, 66-93.

73

70

67

69

DP No. Published as

SPRC Reprint No. (if applicable)

- 39. Bruce Bradbury (1993), 'Unemployment, and Income Support: Challenges for the Years Ahead', *Economic Papers*, Vol. 12, No. 2, June, 14-31.
- 40. Peter Saunders (1993), 'Married Women's Earnings and Family Income Inequality in the Eighties', *Australian Bulletin of Labour*, Vol. 19, No. 3, 3-22.
- 44. Sheila Shaver and Jonathan Bradshaw (1995), 'The Recognition of Wifely Labour by Welfare States', *Social Policy and Administration*, Vol. 29, No.1, March, 10-25.
- 46. Peter Saunders, Björn Halleröd and George Matheson (1994), 'Making Ends Meet in Australia and Sweden: A Comparative Analysis Using the Subjective Poverty Line Methodology', *Acta Sociologica*, Vol. 37, No. 3, 3-22.
- 48. Peter Saunders (1993), 'Poverty and Inequality: Social Security in the 1990s', in J. Disney and L. Briggs, eds, *Social Security Policy: Issues and Options*, papers from the Conference, 'Social Security Policy: The Future', November, AGPS 29-48.
- 49. Peter Saunders (1994), 'Rising on the Tasman Tide: Income Inequality in Australia and New Zealand', *Social Policy Journal of New Zealand*, Issue 2, July, 97-114.
- 52. Peter Saunders, 'The Immigrant Dimension of Income Inequality' in J. Neville, ed., As the Rich Get Richer: Changes in Income Distribution, Committee for the Economic Development of Australia (CEDA), Sydney, 66-86.
- 53. Peter Saunders (1995), 'In Defence of a Poverty Line', *Just Policy*, No. 4, September, 9-16.
- 54. Peter Whiteford (1995), 'The Use of Replacement Rates in International Comparisons of Benefit Systems', *International Social Security Review*, Vol. 48, No.2/95, 3-30.
- 55. Peter Saunders (1995), 'Conditionality and Transition as Issues in the Basic Income Debate', in Income Support in an Open Economy: Basic Income Seminar, Victorian Council of Social Service and the Good Shepherd Youth and Family Services, Melbourne, 51-62.
- 56. Peter Saunders (1995), 'Improving Work Incentives in a Means-Tested Welfare System: The 1994 Australian Social Security Reforms, *Fiscal Studies*, Vol. 16, No. 2, May, 145-70.

DP No. Published as

SPRC Reprint No. (if applicable)

- 60. Johan Fritzell and Peter Saunders (1995), 'Wage and Income Inequality in Two Welfare States: Australia and Sweden', in F. Engelstad, R. Kalleberg, A. Lura and L. MjØset, eds, *Comparative Social Research, Volume 15: Institutional Aspects of Work and Wage Determination, JAI Press,* Greenwich, CT, 187-229. Also in *Comparative Social Research Yearbook*
- 62. Peter Saunders and Michael Fine (1997), 'Evaluation and Research in Social Policy', *Australian Journal of Social Research*, Vol. 3, No. 1, January, 75-94.
- 63. Peter Saunders (1996), 'Unpacking Inequality: Wage Incomes, Disposable Incomes and Living Standards', in *The Industry Commission Conference on Equity, Efficiency and Welfare, Conference Proceedings*, AGPS, Canberra, 225-55.
- 64. Peter Saunders (1996), 'Poverty in the 1990s: A Challenge to Work and Welfare', in P. Sheehan, B. Grewal and M. Kumnick, eds, *Dialogues in Australia's Future: In Honour of the Late Professor Ronald Henderson*, Centre for Strategic Economic Studies, Victoria University of Technology, Melbourne, 325-50.
- 65. Sheila Shaver and Michael Fine (1996), 'Social Policy and Personal Life: Changes in State, Family and Community in the Support of Informal Care' in Aged and Community Care Division and Office of Disability, Department of Human Services and Health, *Towards a National Agenda for Carers, Workshop Papers*, No. 22, AGPS, Canberra, 19-36.